Author's response to reviews

Title: A 12-week after-school physical activity improves endothelial cell function in overweight and obese children: a randomised controlled study

Authors:

Jong-Hwan Park (prof.parkjh@fuji.waseda.jp)
Masashi Miyashita (m.miyashita@aoni.waseda.jp)
Yoo-Chan Kwon (kwonyc@dau.ac.kr)
Hyun-Tae Park (tonypark@ncgg.go.jp)
Eun-Hee Kim (ehk1959@dau.ac.kr)
Jin-Kee Park (park7166@yahoo.co.kr)
Ki-Beam Park (welzel@dau.ac.kr)
Suk-Ran Yoon (sryoon@kribb.re.kr)
Jin-Woong Chung (jwchung@dau.ac.kr)
Yoshio Nakamura (nakamura@waseda.jp)
Sang-Kab Park (sgpark@dau.ac.kr)

Version: 2 Date: 27 February 2012

Author's response to reviews: see over
Dear Melika Redolosa

We greatly appreciate your comments. In accordance with your suggestions, we have revised the manuscript as described below:

1. Please can you name the ethics committee within the manuscript file.

   **Answer 1:** Thank you very much for your comments. This study was approved by the ethics committee of Dong-A University. Therefore, we have made the following change (P.7, Line 8):
   
   “This study was approved by the ethics committee of Dong-A University”

2. Trial Registration Number. We notice that you are reporting a clinical trial but have not cited a trial registration number. This must be obtained before we can begin peer review of your manuscript.

   **Answer 2:** Thank you very much for this comment. We have obtained the Trial Registration Number, which is ISRCTN19037201. We have added this Trial Registration Number into Page 3.

   “Trial registration: Current Controlled Trials ISRCTN19037201”

We very much hope that you find these adjustments satisfactory and any correspondence should be directed to me at the following address:

Sang-Kab Park, PhD.
Dong-A University, College of Sport Sciences
840 Hadan 2-dong, Saha-gu, Busan, 604-714, South Korea
Phone & Fax: +82-51-200-7843
Email: sgpark@dau.ac.kr

Many thanks in advance for considering our paper.

Yours truly,
Sang-Kab Park